Author's response to reviews

Title: Effects of Mycophenolate Mofetil on Key Pattern of Coronary Restenosis: a Cascade of in vitro and ex vivo Models

Authors:

Rainer Voisard (rainer.voisard@medizin.uni-ulm.de)
Sandra Viola (sandra.viola@medizin.uni-ulm.de)
Verena Kaspar (verena.kaspar@medizin.uni-ulm.de)
Christian M Weber (christian.weber@medizin.uni-ulm.de)
Lutz von Muller (lutz.muller@medizin.uni-ulm.de)
Regine Baur (regine.baur@medizin.uni-ulm.de)
Iris Gastrock-Balitsch (iris.gastrock-balitsch@medizin.uni-ulm.de)
Vinzenz Hombach (vinzenz.hombach@medizin.uni-ulm.de)

Version: 4 Date: 2 May 2005

Author's response to reviews:

Ulm, Germany, may 2nd, 2005

Dear BMC Editorial Team,

enclosed please find the second revision II of our manuscript entitled: "Effects of Mycophenolate Mofetil on Key Pattern of Coronary Restenosis: a Cascade of in vitro and ex vivo Models".

Please find here the point by point descriptions of the changes made:

1. Authors that have contributed equally are stated on the title page.
2. Title:
3. Affiliations were adapted.
4. Brackets were removed.
5. The abstract subtitles were adapted.
6. In the references "et al." was removed and all the authors names were included.
7. Figure titles were removed.
8. Figures were transformed to powerpoint.
9. Additional files were removed.

We hope that the manuscript can now be accepted for publication in BMC.

With best wishes

Priv.-Doz. Dr. Rainer Voisard